Author's response to reviews

Title: Abdominal wall hemorrhage after intravenous thrombolysis for acute ischemic stroke

Authors:

Se-A An (ansea@chamc.co.kr)
Jinkwon Kim (kimjinkwon@chamc.co.kr)
Sang Heum Kim (shss96@cha.ac.kr)
Won Chan Kim (wckim@cha.ac.kr)

Version: 5 Date: 18 November 2012

Author's response to reviews: see over
Deesha Majithia  
BMC Neurology  
19th November 2012  

Re: MS: 3900404977952309  
Manuscript Title: Abdominal wall hemorrhage after intravenous thrombolysis for acute ischemic stroke  

Dear Dr. Deesha Majithia,  

Thank you very much for your consideration of our manuscript and request for a revised version. Please find attached revised manuscript entitled “Abdominal wall hemorrhage after intravenous thrombolysis for acute ischemic stroke”, to be considered for publication as a case report in BMC neurology.

We have responded to all the addressed concerns of the reviewers point by point. As you will see, we have made every attempt to incorporate the suggestions as thoroughly as possibly by modifying the sentences and words in the manuscript. The revised or corrected portions in the revised manuscript were underlined.

We wish to thank you again for the reconsideration of our manuscript. Should you have any questions or there are any issues concerning the revised manuscript, please do not hesitate to contact us should

Yours Sincerely,  

Won Chan Kim, M.D.  
Department of Neurology  
CHA Bundang Medical Center  
CHA University College of Medicine  
351 Yatap-dong, Bundang-gu,  
Seongnam 463-712, Republic of Korea  
Tel: +82-31-780-5482  
Fax: +82-31-780-5269  
E-mail: wckim@cha.ac.kr